## **Author's Response To Reviewer Comments**

Clo<u>s</u>e

## Dear Hans,

Here is the manuscript updated to include the Oxford Nanopore sequence accession numbers (line 175), the updated BUSCO results (lines 321-328), and the GigaDB reference in the "Availability of supporting data" section (line 410).

Regards, Ben

Clo<u>s</u>e